Author’s response to reviews

Title: Internal mammary artery dilatation in a patient with aortic coarctation, aortic stenosis, and coronary disease. Case report

Authors:

Jose Rubio Alvarez (jose.rubio.alvarez@sergas.es)
Laura Reija Lopez (framan1@hotmail.com)
Juan Sierra Quiroga (juan.sierra.quiroga@sergas.es)
Jose M Martinez Comendador (jose.manuel.martinez.comendador@sergas.es)
Anxo Martinez de Alegria (anxo.martinez.dealegria@sergas.es)
Jose M Martinez Cereijo (jose.manuel.martinez.cereijo@sergas.es)
Cristian Delgado Dominguez (cristian.delgado.dominguez@sergas.es)

Version: 5 Date: 4 March 2011

Author’s response to reviews:

The list of authors in the submission system were modified.
The identifying marks were removed from the figures 3 and 4.